Title: Cluster randomized controlled trial protocol: Addressing Reproductive Coercion in Health Settings (ARCHES)

Authors:

Daniel J Tancredi PhD (djtancredi@ucdavis.edu)
Jay G Silverman PhD (jgsilverman@ucsd.edu)
Michele R Decker ScD, MPH (mdecker@jhsp.h.edu)
Heather L McCauley ScD, MS (heather.mccauley@chp.edu)
Heather Anderson (heather.anderson@chp.edu)
Kelley A Jones MPH (kaj25@pitt.edu)
Samantha Ciaravino (ciaravinos@upmc.edu)
Angela Hicks (angela.hicks@chp.edu)
Claire Raible (claire.raible@chp.edu)
Sarah Zelazny (sarah.zelazny@chp.edu)
Lisa James MS (ljames@futureswithoutviolence.org)
Elizabeth Miller MD, PhD (elizabeth.miller@chp.edu)

Version: 3
Date: 17 January 2015

Author’s response to reviews:

The slight revisions on December 10, 2014 from the original version (submitted December 9, 2014) were to remove a stray copy of a figure on page 12 and a stray figure caption on page 20 of the manuscript text, vestiges from when the figures were extracted from the manuscript and placed into separate files. The revisions today (January 17, 2015) were to clarify and correct the description of the stratified randomization procedures used to ensure balance in the number of urban and rural clinics assigned to the two study arms.